Author's response to reviews

Title: Pro-neural transcription factors as cancer markers

Authors:

Maria Vias (maria.vias@cancer.org.uk)
Charles E. Massie (charlie.massie@cancer.org.uk)
Philip East (phil.east@cancer.org.uk)
Helen Scott (helen.scott@cancer.org.uk)
Anne Warren (ayw23@cam.ac.uk)
Zongxiang Zhou (zz34@cornell.edu)
Alexander Yu. Nikitin (an58@cornell.edu)
David E. Neal (den22@cam.ac.uk)
Ian G. Mills (ian.mills@cancer.org.uk)

Version: 5 Date: 8 May 2008

Author's response to reviews: see over
8/05/08

Dear Editor,

Many thanks for your further feedback. I have removed boxed text with the exception of a primer table which is now Table 3 at the end of the manuscript. R script information has been deleted from the text and is now provided as figures 1 and 8. Figures have been renumbered accordingly and references to the two new figures inserted into the text. The order in which the figures are referred to in the text has been rechecked and corrected. The table titles and legends have been separated as requested. I would be grateful if you could give me an indication of the likely publication date.

Many thanks and best wishes,

Ian Mills.

-----Original Message-----
From: Core pre-accept [mailto:corepreaccept@biomedcentral.com]
Sent: Thu 2008-05-08 11:56
To: Ian Mills
Subject: Your manuscript 1991865293174656 requires further formatting changes

Authors: Maria Vias, Charles E. Massie, Philip East, Helen Scott, Anne Warren, Zongxiang Zhou, Alexander Yu. Nikitin, David E. Neal and Ian G. Mills
Title: Pro-neural transcription factors as cancer markers
Journal: BMC Medical Genomics
MS: 1991865293174656

Dear Dr Mills,

Thank you very much for submitting the revised version of your manuscript. Unfortunately there are still some changes that need to be made prior to publication. Can you please carry out the additional revisions listed and complete a new submission for your manuscript. The instructions to submit your revised manuscript are available below.

Please ensure that all revisions are made to the most recent version of your article which can be accessed at the following link:
http://www.biomedcentral.com/media/man/1991865293174656/msd/main/0

Figures:
http://www.biomedcentral.com/media/man/1991865293174656/msd/FIG/1
http://www.biomedcentral.com/media/man/1991865293174656/msd/FIG/2
http://www.biomedcentral.com/media/man/1991865293174656/msd/FIG/3
http://www.biomedcentral.com/media/man/1991865293174656/msd/FIG/4
http://www.biomedcentral.com/media/man/1991865293174656/msd/FIG/5
Formatting changes requested
-------------------------------

Boxes - You have included boxed text in the text of the manuscript. Unfortunately due to the format of the on-line manuscript we cannot incorporate these. Please either include the text within the manuscript, or move these to the end of the manuscript with the tables. Please remember to up-date any references within the text.

R script information - Please delete this from the text and include it in a figure file. Please remember to include a legend for the figure at the end of the manuscript and to include a reference to the figure in the text.

Figure order - You currently have figures 5 and 7 mentioned before the other figures. The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. The figures should be provided in this same order through the submission system. Every figure provided must be referred to in the text.

Table title and legend - The table title should appear above the table. The legend (or key) should appear below the table.

Thank You
-------------------------------

To submit your revised manuscript
-------------------------------

1. Go to 'my BioMed Central' (http://www.biomedcentral.com/my/) and log on with your email address and password. Then click on 'My manuscripts'. You will reach a page giving details of all the manuscripts you have submitted to journals published by BioMed Central.
2. Click on the 'Submit revision' button next to the title of this manuscript to go through to a submission form.
3. With the 'Manuscript details' tab, please update the title, abstract and author details if they have changed since the previous version. It is very important that all changes are updated on this page, as well as in the manuscript file as the information on this page will be used in PubMed and on BioMed Central.
4. With the 'Cover letter' tab, please provide a covering letter with a point-by-point description of the changes made.
5. With the 'Upload files' tab, please upload the revised version of the manuscript and press 'Submit new version'. Please wait for the confirmation page to appear - this may take a few moments. Please do not hesitate to contact us if you have any problems or questions.

With all best wishes,

The BioMed Central Editorial Production Team